**Author’s response to reviews**

**Title:** The association between delusional-like experiences, and tobacco, alcohol or cannabis use: a nationwide population-based survey.

**Authors:**
- Sukanta Saha (kristin.laurens@unsw.edu.au)
- James G Scott (James_G_Scott@health.qld.gov.au)
- Daniel Varghese (Daniel_Varghese@health.qld.gov.au)
- Louisa Degenhardt (louisa@burnet.edu.au)
- Tim Slade (tims@unsw.edu.au)
- John J McGrath (John_mcgrath@qcmhr.uq.edu.au)

**Version:** 2  **Date:** 9 October 2011

**Author’s response to reviews:**

Dear Colleague,

The required text related to ethics and consent has been included.

Kind regards,

John McGrath